Author's response to reviews

Title: Season of birth is different in Inuit suicide victims born into Traditional than into Modern Lifestyle. A register study from Greenland.

Authors:

Karin S Björkstén (Karin.Sparring.Bjorksten@ki.se)
Peter Bjerregaard (pb@si-folkesundhed.Dk)

Version: 3 Date: 16 January 2015

Author's response to reviews:

Dr. Wouter van Ballegooijen

Dear Editor,

We thank you and the reviewers for reviewing our paper

MS: 3828246721446532 Season of Birth vary in Inuit Suicide Victims born into traditional but not into modern Lifestyle. A Register Study from Greenland by Karin S Björkstén and Peter Bjerregaard

We have read the reviewers' comments carefully and used most of them to clarify questions and doubts and make it more reader-friendly.

Please find enclosed the answers to the reviewers' questions are written in red.

The reviewers had many questions about Greenland, so we wrote a long part about Greenland in the Methods section.

We re-organised the manuscript in several ways to make it easier to understand.

Since the subject is complicated, we introduced several sub-headings.

There are some new references according to suggestions.

A language editor helped us with the new version.

We changed the title to

Season of birth is different in Inuit suicide victims born into Traditional than into Modern Lifestyle. A register study from Greenland.

We now believe that our paper is easier to read and understand.

Karin Sparring Björkstén